Author's response to reviews

Title: Development of AMSTAR: a measurement tool to assess the methodological quality of systematic reviews

Authors:

Beverley J Shea (bevshea@CIET.org)
Jeremy M Grimshaw (jgrimshaw@ohri.ca)
George A Wells (gawells@ottawaheart.ca)
Maarten Boers (mboers@vumc.nl)
Neil Andersson (CIETinter@compuserve.com)
Candyce Hamel (chamel@uottawa.ca)
Ashley C Porter (ashleyclaireporter@hotmail.com)
Peter Tugwell (ptugwell@uottawa.ca)
David Moher (dmoher@uottawa.ca)
Lex M Bouter (lm.bouter@dienst.vu.nl)

Version: 4 Date: 14 February 2007

Author's response to reviews:

February 14, 2007

Dear BioMed Central Editorial Team,

Thank you for the opportunity to publish our manuscript entitled 'Development of AMSTAR: a measurement tool to assess the methodological quality of systematic reviews.'

We have made changes based on the suggestions received from your publishing group. You will find a revised version of the manuscript. If you have further comments, please, do not hesitate to contact me at anytime.

Best wishes,
Beverley Shea